Author's response to reviews

Title: Women with learning disabilities and Read coding: Lessons from a cohort study

Authors:

Fiona Reynolds (FionaReynolds@gmail.com)
Debbi L Stanistreet (dlstanistreet@liverpool.ac.uk)

Version: 6 Date: 19 July 2008

Author's response to reviews:

Reference 14 - We are unable to references as 14a or 14b. Please make sure references are cited in ascending numerical order.

This has been amended.

Tables - Please remove the tables from the text of the manuscript. These should be included after the reference list only.

This has been done.